Author's response to reviews

Title: Radiation Therapy for Desmoplastic Medulloblastoma - A Retrospective Analysis of Outcome and Prognostic Factors

Authors:

Stefan Rieken (srieken@arcor.de)
Timo Gaiser (timo.gaiser@med.uni-heidelberg.de)
Angela Mohr (angela.mohr@med.uni-heidelberg.de)
Thomas Welzel (thomas-welzel@med.uni-heidelberg.de)
Olaf Witt (olaf.witt@med.uni-heidelberg.de)
Andreas E. Kulozik (andreas.kulozik@med.uni-heidelberg.de)
Wolfgang Wick (wolfgang.wick@med.uni-heidelberg.de)
Jürgen Debus (juerger.debus@med.uni-heidelberg.de)
Stephanie E. Combs (stephanie.combs@med.uni-heidelberg.de)

Version: 6 Date: 2 August 2010

Author's response to reviews: see over
Dear Ms Neilan,
dear Ms. Cassady-Cain,

today we are submitting our re-re-re-revised original article entitled

**Outcome and Prognostic Factors of Desmoplastic Medulloblastoma treated within a Multidisciplinary Treatment Concept**

We added a comment to the manuscript stating that in our institute no official ethical approval must be obtained for retrospective and anonymously performed analyses.

We adapted citation no. 33 now referring to Karger’s “Onkologie”, where all abstracts from the conference DKK2010 are published (Vol. 33, Suppl. 2).

All changes to the manuscript are highlighted in red color.

We hope that our manuscript is now suitable for publication in BMC Cancer and look forward to hearing from you soon.

Sincerely

Dr. Stephanie Combs and Dr. Stefan Rieken
For all co-authors